Author's response to reviews

Title: Anti-hLAMP2-antibodies and dual positivity for anti-GBM and MPO-ANCA in a patient with relapsing pulmonary-renal syndrome.

Authors:

Christoph Etter (christoph.etter@web.de)
Ariana Gaspert (ariana.gaspert@usz.ch)
Stephan Regenass (stephan.regenass@usz.ch)
Rudolf P. Wuethrich (rudolf.wuethrich@usz.ch)
Thomas Kistler (thomas.kistler@ksw.ch)
Renate Kain (renate.kain@meduniwien.ac.at)
Clemens D. Cohen (clemens.cohen@usz.ch)

Version: 4 Date: 8 June 2011

Author's response to reviews: see over
Dear Mrs Chap

Thank you for accepting our revised manuscript “Anti-hLAMP2-antibodies and dual positivity for anti-GBM and MPO-ANCA in a patient with relapsing pulmonary-renal syndrome” for publication. Please find enclosed the third revision of the manuscript.

Sincerely,
Christoph Etter and Clemens D. Cohen

- Abstract: this abstract does not have the correct structure. Please reformat it to use these headings: Background, Case Presentation, Conclusions.
done

- Authors’ contributions: Please indicate that all authors read and approved the final manuscript.
Authors read and approved the final manuscript. This is indicated in the corresponding section.
done

- Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.
done

- References: Please make sure references are cited in numerical order. Reference 16 is cited before 14 and 15. Please change.
References are now in numerical order.
done

- Tables: we are unable to display shading and colours within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold.
done

- Email address: please include email addresses for all the authors on the title page.
done

- Tables- please remove the visible vertical lines from your tables
done

- Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions.
done